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TABLE 4. %IZES, Rr':vowno»és PER MINUTE, HEADS, POWER REQUIRED AND BOILER Continuously Operating Return Pumps
o ORSEPOWER FOR SEVERAL CO AL AL : . : . : o
‘ COMMERCIAL S1zES OF CENTRIFUG The piston speed in feet per minute should not be more than 10 times

Tyre BoiLer Feep P h A
D s the square root of the number of inches in the length of stroke.

; smh:,, por | Buurs - Preg Sumes, INcmzs. Caracrrr Bouzs Houssrowss The dnormal capacity of stand_afr,d centrifugal C1l)um})shsho1\;ld be based
CaES ORSEPOWER - ALLONE RVED WiTHOUT on condensate at a temperature of not over 180 deg. fahr. For tempera-
i ree M Sarerr F. . : . I 4 : N
, Suction l Discharge o prmTY Hhoon , tures of condensate above this the capacity should be increased as in
Table 6. . ’
Two Stage for 100 ib. Working Pressure A The increase in pump capacity may be reduced if a static head above L
22,% }gg 214 2. 100 1,450 . TasLE 6. CORRECTION FACTORS FOR STANDARD CENTRIFUGAL PuMPS FOR TEMPERA-
2 162 i ) g% g(s) 2,175 TURES OF CONDENSATE ABOVE 180 DEG. Farr. AT PuMp SucTtioN i
. 3,262
-4 7.2 e i ; i
5 ‘27‘0 ‘75 g . é(z)g ) gv% Dxg. Fabr, - Facror
6 76.5 8 6 900 13,000 190 L.15
: 200 : 1.56
} 204 : 2.00
Three Stage for 150 Ib. Working Pressure ¢
. . Note.—To use this Table, multiply the quantity of condensate to be handled by the factor corresponding
2 9.9 ’ VZI/ i ™ to the temperature of the condensate at the pump suction and select a pump for the quantity thus found.
- 2 2 100 1,450 : o »
i ‘gg ; ‘é i 225 3,262 - TaBLE 7. SiMPLEX P1sToN TyPE STEAM DRIVEN CONDENSATION PUMP AND RECEIVERS
E 8.3 4 5 ggg g:ggg ’ CrunpER 8 . Pire Sizes
.6 11‘%.7 8 6 900 . 13,000 lucm{zm . TNcrEs DeamniNg Caraciry
- Y B
: Steam | Water | Stroke | Stesm | Exhaust | Inlet | Delivery | Square Feet 1Ix Pree
Four Stage for 200 ib. Working Pressure : : K T )
2 33.2 2% 2 100 1,450 | 1. 2/é ;/4 3% | 2 ;//2 1,000 3,000
1 . : 3 225 3.262 3 134 23 B4 14 2 4 . 6,000
93.0 6 4 400 5800 4 2% | 3 % | % | 2 3% 3,000 9,000
5 143.0 7 5 620 - 3'990 5 3 314 1g %1 2 1 4,500 13,500
6 192.0 8 6 900 - 13000 -, gy ¢ 3//2 RO i% 7300 27000
- 2 2 4 ’ »
6 T 3% 8 34 1 2%% 2 13,000 39,000
: 6 4 8 34 - 1 214 2 15,000 45,000
TABLE 5. ScrRew Pumes 7 4 8 1 1Y 244 2 15,000 45,000
- 5 " 10 1 1 3 2 27,000. 81,000
- Suz or Poue Garcons Maxmon s : 1(8) "6 12 1Y 1%/2 ) 4% 3% 40,000 | 120,000
Tenzs vsa Movora RevoLvmons Tncams - Dpomases 12 M| 12 || 2 5 4 65.000 | 195,000
B 12 8 12 1% 2 5 4 70,000 210,000
2 ' 2- 15 1,600 2 L
SV2 28: §8 }"% . %1/ éif/é o the pump suction is provided. When this static head is made equivalent
3 40- 60 1,600 37 2}};’ T to-the absolute boiling pressure of the condensate (measured in feet of
_ i% ~8555: ;gg 1,500 4 3 ‘water) minus 12 ft., no increase in pump capacity is required.
5 . 175-1275 | i'% g i This static suction head requirement may also be reduced about 50
g% . %(7)0— 325 1,200 5 4 per cent when special large suction inlets or special impellers are provided.
5- 475 1,200 6 6 - Sufficient head in additi “th I head '
7 ’ . . ufficient head in addition to’the total head necessary to overcome
8 4300—50_ ?(5)8 l.g?g _ g _ g . static head, velocity head, pipe friction and boiler pressure should-be
1(% 700-1000 720 10 10 provided wherever condensate is to be returned directly to a boiler from
800-1400 700 12 12 the pump. '
12 1200-2100 600 14 ) . .
16 1750-4200 425 16 - H . _ VACUUM HEATING PUMPS
' - ' Return-line vacuum heating pumps may be of the following types:
312 o : N -
: .o 313




